Author's response to reviews

Title: Which health care facilities do adult Malawian antiretroviral therapy patients utilize during intercurrent illness? A cross sectional study

Authors:

Caroline Masangalawe (cmasangalawe@medcol.mw)
Akuzike Kandulu (akandulu@medcol.mw)
Joep J van Oosterhout (joepvanoosterhout@gmail.com)

Version: 6 Date: 14 December 2011

Author's response to reviews: see over
Dear Editors,

On behalf of my fellow authors I am submitting the R2 version of our manuscript entitled:

**WHICH HEALTH CARE FACILITIES DO ADULT MALAWIAN ANTIRETROVIRAL THERAPY PATIENTS UTILIZE DURING INTERCURRENT ILLNESS? A CROSS SECTIONAL STUDY**

We thank editors and reviewers for accepting our manuscript for publication in BMC Health Services Research.

We have made an adjustment to the 4th sentence of the methods section as recommended by reviewer 1, which was the only comment of reviewers on the R1 version of the manuscript.

Best regards,

Joep J van Oosterhout, PhD
Associate Professor
Department of Medicine, College of Medicine
Blantyre, Malawi